Author's response to reviews

Title: Antimicrobial potential of the Ethiopian Thymus schimperi essential oil against Dermatophyte spp. and other Micro organisms

Authors:

Mohammmed N Nassir (mohamicro2008@yahoo.com)
Ketema T Tafess (ttafess@gmail.com)
Dawit Abate A Abate (dr.dawitabate@yahoo.com)

Version: 2 Date: 7 January 2015

Author's response to reviews:

Change included
We have included the author's information including their email address on the first page of the main manuscript
We have also number the main manuscript